 Vc/9-Aug-04
Week33 starting 9Aug  04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 09 aug04  ~17.45 hrs 



	TBA1
	MBAR1139
	PT 10 Min. NRJ 11850 A
	1.9

	TBA2
	MBBL2089
	Prep 5 Pumping + Cool Down
	1.94

	TBB1
	SSS041
	PT 6.2 Training 2
	3.3 HV

	TBB2
	MBBL1121
	Prep 5 Pumping + Cool Down
	302.5

	TBC1
	MBBL3194
	Prep 5 Pumping + Cool Down
	89

	TBC2
	MBBL2108
	PT 12 Warm Up
	300.8

	TBD1
	SSS007
	Stretched Wire Warm SW13
	301.4

	TBD2
	SSS036
	ICS 5 Traveller Last
	301.9

	TBE1
	MBBL3193
	ICS 5 Traveller Last
	302.5

	TBE2
	MBAL3113
	ICS 3 Traveller after warm test
	298

	TBF1
	MBAL2047
	PT 3 QH Measure
	1.9

	TBF2
	MBBL1120
	PT 6.4 Training 4
	1.9


8/9/2004 5:40:25 PM    Temperature data are taken from TBSM18 site
Coordinator this week: Gijs; MTM support on call :  (16-3602).

12 Magnets( 11 dipoles+ 1 SSS)   Disconnected after tests last week    ( Sat Midnight to Sat midnight)  ; 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.(Jerome ?)

· SSS work :
· RESISTANCE M’ments at warm at the beginning & end 
1. Orbital corrector dipoles

2. Cryo heater

3. Temp Sensor TT 821

 

Item 2 & 3 can be included without any trouble

Howver for item 1 the availability of current source 10 mA, 10 VDC is Required ( Tcoil supply ?)

Bench status  and tests for now( Monday 2 Aug/7at 17.40 hrs)  & next few days :  see Video Display

 

	Magnets under Test on 09-08 5:46:25 PM 

	-
	Bench
	Magnet
	Shafts
	Start
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	Last Test
Desc
	
Time
	
I
	
Temp.
	Total

	1
	TBA1
	MBAR1139 
	[image: image1.png]



	02-08 
	50 h.
	106 h.
	25 h.
	-
	-
	PT 10 Min. NRJ 11850 A
	09-08
(16 : 56) 
	11850 A. 
	1.9 K
	181 hours

	2
	TBA2
	MBBL2089 
	[image: image2.png]



	06-08 
	27 h.
	44 h.
	-
	-
	-
	Prep 5 Pumping + Cool Down
	07-08
(21 : 30) 
	A. 
	Cooling
	71 hours

	3
	TBB1
	SSS041 
	[image: image3.png]



	02-08 
	132 h.
	32 h.
	11 h.
	-
	-
	PT 6.2 Training 2
	09-08
(16 : 56) 
	12610 A. 
	1.9 K
	175 hours

	4
	TBB2
	MBBL1121 
	[image: image4.png]



	05-08 
	48 h.
	44 h.
	-
	-
	-
	Prep 5 Pumping + Cool Down
	07-08
(21 : 31) 
	A. 
	Cooling
	92 hours

	5
	TBC1
	MBBL3194 
	[image: image5.png]



	07-08 
	28 h.
	15 h.
	-
	-
	-
	Prep 5 Pumping + Cool Down
	09-08
(02 : 42) 
	2 A. 
	Cooling
	43 hours

	6
	TBC2
	MBBL2108 
	[image: image6.png]



	04-08 
	44 h.
	46 h.
	20 h.
	15 h.
	-
	PT 12 Warm Up
	09-08
(02 : 30) 
	2 A. 
	Warming
	125 hours

	7
	TBD1
	SSS007 
	[image: image7.png]



	05-08 
	116 h.
	-
	-
	-
	-
	Stretched Wire Warm SW13
	08-08
(01 : 52) 
	0 A. 
	300 K
	116 hours

	8
	TBD2
	SSS036 
	[image: image8.png]



	28-07 
	156 h.
	4 h.
	63 h.
	22 h.
	2 h.
	ICS 5 Traveller Last
	08-08
(01 : 38) 
	0 A. 
	300 K
	287 hours

	9
	TBE1
	MBBL3193 
	[image: image9.png]



	04-08 
	58 h.
	16 h.
	73 h.
	-
	-
	ICS 5 Traveller Last
	09-08
(10 : 30) 
	0 A. 
	300 K
	147 hours

	10
	TBE2
	MBAL3113 
	[image: image10.png]



	07-08 
	46 h.
	-
	-
	-
	-
	ICS 3 Traveller after warm test
	08-08
(01 : 30) 
	0 A. 
	300 K
	46 hours

	11
	TBF1
	MBAL2047 
	[image: image11.png]



	04-08 
	40 h.
	92 h.
	10 h.
	-
	-
	PT 3 QH Measure
	09-08
(09 : 07) 
	0 A. 
	1.9 K
	142 hours

	12
	TBF2
	MBBL1120 
	[image: image12.png]



	05-08 
	49 h.
	7 h.
	37 h.
	-
	-
	PT 6.4 Training 4
	09-08
(14 : 22) 
	12437 A. 
	1.9 K
	93 hours


Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAR1139
	11634 Amp.
(D1-L)
	12507 Amp.
(D2-L)
	[image: image13.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL2089
	[image: image14.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS041
	12281 Amp.
(QI-12)
	12610 Amp.
(No Data)
	[image: image15.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBBL1121
	[image: image16.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBBL3194
	[image: image17.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBBL2108
	10420 Amp.
(D1-L)
	12404 Amp.
(D2-L)
	[image: image18.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS007
	[image: image19.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS036
	12482 Amp.
(Q_I_12)
	12491 Amp.
(Q_I_12)
	12850 Amp.
(Q-E-34)
	[image: image20.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBL3193
	9536 Amp.
(D2-U)
	10352 Amp.
(D1_U)
	11515 Amp.
(D2_U)
	11490 Amp.
(D2_U)
	11809 Amp.
(D2-U)
	12038 Amp.
(D2_L)
	12143 Amp.
(D2-U)
	12401 Amp.
(D2-U)
	12659 Amp.
(D2_U)
	[image: image21.jpg]



	-
	-
	-

	10
	MBAL3113
	[image: image22.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAL2047
	[image: image23.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBL1120
	12012 Amp.
(D2-U-S13)
	11901 Amp.
(D1_L_S13)
	12222 Amp.
(D2-L-S13)
	12437 Amp.
(D2-U-S13)
	[image: image24.jpg]



	-
	-
	-
	-
	-
	-
	-
	-


Results by Week for PREVIOUS week 

	Magnets under Test from 01-08-2004 to 07-08-2004 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBAL3177 
	[image: image25.png]



	25-07-2004 
	37 h.
	50 h.
	71 h.
	15 h.
	15 h.
	02-08-2004 
	192 hours
	
	

	2
	TBA2
	MBBL3191 
	[image: image26.png]



	31-07-2004 
	34 h.
	55 h.
	37 h.
	17 h.
	8 h.
	06-08-2004 
	144 hours
	
	

	3
	TBB1
	MBBL3190 
	[image: image27.png]



	26-07-2004 
	33 h.
	48 h.
	36 h.
	17 h.
	6 h.
	02-08-2004 
	168 hours
	
	

	4
	TBB2
	MBBL2090 
	[image: image28.png]



	29-07-2004 
	30 h.
	52 h.
	66 h.
	17 h.
	8 h.
	05-08-2004 
	168 hours
	
	

	5
	TBC1
	MBAL3110 
	[image: image29.png]



	03-08-2004 
	30 h.
	41 h.
	26 h.
	10 h.
	10 h.
	07-08-2004 
	96 hours
	
	

	6
	TBC1
	MBAL3172 
	[image: image30.png]



	26-07-2004 
	48 h.
	71 h.
	34 h.
	20 h.
	12 h.
	03-08-2004 
	192 hours
	
	

	7
	TBD1
	SSS023 
	[image: image31.png]



	19-07-2004 
	98 h.
	95 h.
	119 h.
	17 h.
	1 h.
	03-08-2004 
	360 hours
	
	

	8
	TBE1
	MBBL3137 
	[image: image32.png]



	28-07-2004 
	58 h.
	29 h.
	41 h.
	16 h.
	12 h.
	03-08-2004 
	144 hours
	
	

	9
	TBE2
	MBAL3170 
	[image: image33.png]



	02-08-2004 
	9 h.
	51 h.
	23 h.
	18 h.
	2 h.
	07-08-2004 
	120 hours
	
	

	10
	TBE2
	MBBL2035 
	[image: image34.png]



	23-07-2004 
	37 h.
	114 h.
	48 h.
	23 h.
	7 h.
	02-08-2004 
	240 hours
	
	

	11
	TBF1
	MBAR1140 
	[image: image35.png]



	27-07-2004 
	24 h.
	47 h.
	78 h.
	13 h.
	6 h.
	03-08-2004 
	168 hours
	
	

	12
	TBF2
	MBBL1119 
	[image: image36.png]



	29-07-2004 
	33 h.
	67 h.
	44 h.
	13 h.
	8 h.
	05-08-2004 
	168 hours
	
	


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL3177
	11627 A.
(D2-U)
	12226 A.
(D2-U)
	[image: image37.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL3191
	11167 A.
(D1-U)
	12329 A.
(D2-U)
	[image: image38.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBBL3190
	10479 A.
(D2-L)
	12519 A.
(D2_L)
	[image: image39.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBBL2090
	10787 A.
(D1-U-S01)
	11153 A.
(D2 L S01)
	12412 A.
(D1_U S03)
	12816 A.
(D2_L_S12)
	12850 A.
[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBAL3110
	12109 A.
(D2_U)
	12406 A.
(D1_U)
	[image: image41.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBAL3172
	11656 A.
(D2_U)
	12851 A.
[image: image42.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS023
	12287 A.
(QI-34)
	12616 A.
(QI-12)
	12686 A.
(QE_12)
	12845 A.
(QE_12)
	12855 A.
(QE_34)
	[image: image43.png]



	-
	-
	-
	-
	-
	-
	-

	8
	MBBL3137
	11089 A.
(D2-L)
	12296 A.
(D2_U)
	[image: image44.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBAL3170
	12497 A.
(D2-U)
	12642 A.
(D2_U)
	[image: image45.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBBL2035
	11580 A.
(D2-U)
	11859 A.
(D1_U)
	12616 A.
(D1-L)
	12850 A.
[image: image46.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAR1140
	11202 A.
(D2-L)
	11645 A.
(D2-L)
	12142 A.
(D2-L)
	12383 A.
(D1-L)
	12513 A.
(D1-U)
	12597 A.
(D2 L)
	12662 A.
(D1-U)
	12700 A.
(D2-U)
	12332 A.
(D2_U)
	[image: image47.png]



	-
	-
	-

	12
	MBBL1119
	11940 A.
(D2-U-S13)
	12119 A.
(D1-L-S13)
	12231 A.
(D2-U-S13)
	12498 A.
(D2_U-S01)
	12691 A.
(D2-U-S13)
	[image: image48.png]



	-
	-
	-
	-
	-
	-
	-


General Problems last week :  
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _03_A


	TBB2
	VME error during LHC cycle. Measurements stuck in one current plateau.

Same problem on the load line.


	AB-CO
	MMP
	LHC cycle aborted: Load line launched.

Giorgio informed. Origin?

Run recovery performed.

	Wednesday_04_A

Friday_06_N
	TBD2
	HV Test failed on EF circuit (did not pass 800 V) for SSS36

NC remained after the power test.
	MTM
	
	TC informed. NC opened. OK to continue but power test on corrector EF will be performed 

· after the MQ training

· after another HV test on EF 

· if insulation is not degraded.

	Wednesday _04_A


	TBA2
	Training quench one on 3091: 2 DAQ field created and red alarm on QHeaters.

“Heaters should fire after the DAQ trigger”.


	MTM
	AQA
	Check by Giorgio and Stephane using LF DAQ.

Acquisition problem.

Reset the crate and OK to continue

	Wednesday_04_N
	TBE2
	Rotary switch did not work.

Then TEMA blocked on E1 (communication problem between PLCSE  and application.
	MTM
	
	Giorgio informed. He solved the problem

	Thursday_05_M
	TBE2
	HV test failed : TT821 vs ground. Connector inside the IFS box damaged during preparation
	ICS
	
	Giorgio and Kevin repaired the pin which was bent.

	Thursday _05_M
	TBD2
	SPA happened at 5000 A .Giorgio was there: he checked the PLC before resetting: no apparent fault 
	AB/CO?
	
	Origin not known: Communication problem between PLCSE and PC

	Friday_06_N
	TBD2
	TCLaux error on the PLC. T problem on the f auxiliary current leads
	ACR
	
	Giorgio called but it is a cryo problem. Solved by using a potentiometer to keep the valves opened.

	Friday_06_N
	TBD1
	IAP @warm: Vtap signals not visible in IAP but visible in lfoff except IAP_QE_P12.
	MTM
	IAP
	Giorgio informed: He told Nair that the reason is that the SSS is of B type and the corresponding name is IAP_QE_P34. 

	Saturday _07_A
	TBD1
	Axis measurements on SSS07 at warm: physical offsets in stage A and B were above the allowable limits. 

Repeat again: the wire snapped.

Repeat third time: Results not displayed in the screen?
	MTM
	
	Origin not known: Communication problem between PLCSE and PC

	Saturday _07_N
	TBE1
	Training quench Nº9: three files a09, a10, a11. The first 2 are erroneous files. a11 is the real quench. Data could not be found.
	AB/CO
	
	Giorgio comments: the data are present but on the local machine only. Error when copying to storage disk.

	Sunday _08_A
	TBD2
	HV test at warm: Connector C11 broken.
	MTM
	
	Giorgio advised to continue the other steps. Then he came and unbent the pin: then OK.

	Sunday _08_A
	TBF2
	1120 magnet: Provoked quench at 1.5 kA. No trigger generated because no quench. Quench heaters 3&4 in protection did not discharge. Test repeated with 1&2 in protection. Quench appeared.


	MTM
	
	Giorgio informed: he compared with the results of the first run: the same!

As protection is done by HF heaters, the test continued.

Stephane comments: a NC has to be opened and the magnet in stand by in any case.

	Monday _09_M
	TBF2
	SPA occurred during LHC cycle.
	?
	
	Origin unknown.








